Author’s response to reviews

Title: The Effect of Educational Intervention Program on Promoting Preventive Behaviors of Urinary Tract Infection in girls: A Randomized Controlled Trial

Authors:

Zahra Ahmadi (ahmadizahra313@gmail.com)
mohsen shamsi (mohsen_shamsi1360@yahoo.com)
Nasrin Roozbahani (roozbahani7281@gmail.com)
Rahmatalah Moradzadeh (moradzadehr@yahoo.com)

Version: 3 Date: 12 Feb 2020

Author’s response to reviews:

Dear Editorial,

BMC Pediatrics

We would like to appreciate you for giving us the opportunity to revise and resubmit our manuscript to the BMC Pediatrics. We have revised the manuscript according to the editor comments. We thank the editor for their informative comments, and our point-by-point responses to the editor comments are given below.

Manuscript number: BPED-D-19-00885R 2

Full title: The Effect of Educational Intervention Program on Promoting Preventive Behaviors of Urinary Tract Infection in girls: A Randomized Controlled Trial

Editor comment:

- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
Response:

According to the dear editor, the following sentence was revised and edited in 'Ethics Approval and Consent to Participate' section.

Informed written consent was obtained from the mothers and all the procedures performed in the study involving human participants, were in accordance with the ethical standards.

With best wishes